Author’s response to reviews

Title: Severe Generalised Hypersensitivity Reaction to Topical Neomycin after Cataract Surgery - A Case Report I Ansari MBBS, MRCOphth, FRCS (Glasg) Ophthalmologist E Onyema MSc, FRCSEd Ophthalmologist Department of Ophthalmology Princess Alexandra Hospital Harlow, Essex, UK
Correspondence: I Ansari Tel: 44 1279 444455 Fax: 44 827521 E-mail: dociansari@hotmail.com

Authors:

Imran A Ansari (dociansari@hotmail.com)
Ernest Onyema (eyemd121@gmail.com)

Version: 2 Date: 12 August 2007

Author’s response to reviews:

I have made the requested changes which are as below:

- The title page should include the title, authors, author affiliations and corresponding author details.
- Page 2 should contain the abstract (please ensure there are no references cited within the abstract).
- Please insert a Conclusions section after Case presentation.
- Please insert a Competing interests section after the Conclusions. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."
- Please insert an Authors’ contributions section, outlining the role of each author in the study and manuscript preparation.
- Please insert a Consent section after the Conclusions section. The authors should state whether written informed patient consent was obtained for publication.
- Please detail any sources of funding within an Acknowledgements section.
- Please provide Figure titles for each figure and place these in a separate section after the reference list.
- Please further conceal the identity of the patient in Figure 1 - perhaps crop the figure to show only the left eye.